Author’s response to reviews

Title: Case Report: GABAB receptor encephalitis in a patient diagnosed with Amyotrophic Lateral Sclerosis

Authors:
Heike Schumacher (schumacher.heike2@gmail.com)
Thomas Meyer (Thomas.Meyer@charite.de)
Harald Pruess (harald.pruess@dzne.de)

Version: 1 Date: 18 Apr 2018

Author’s response to reviews:

Dear Mrs. Minien,

thank you very much for your revision and your comments.

You can find the edited manuscript, as well as the completed CARE-checklist attached.

Please apologize the confusion concerning the informed consent of publication. The husband of the deceased patient gave us his informed consent (not the patient herself).

We hope that our manuscript now adheres to all editorial requirements.

Yours sincerely,

Heike Schumacher